Allopurinol and cardiovascular outcomes in patients with ischaemic heart disease: the ALL-HEART RCT and economic evaluation

Isla S Mackenzie,^{1*} Christopher J Hawkey,² Ian Ford,³ Nicola Greenlaw,³ Filippo Pigazzani,¹ Amy Rogers,¹ Allan D Struthers,¹ Alan G Begg,¹ Li Wei,⁴ Anthony J Avery,⁵ Jaspal S Taggar,⁵ Andrew Walker,⁶ Suzanne L Duce,¹ Rebecca J Barr,¹ Jennifer S Dumbleton,² Evelien D Rooke,¹ Jonathan N Townend,⁷ Lewis D Ritchie⁸ and Thomas M MacDonald¹ on behalf of the ALL-HEART Study Group

¹MEMO Research, Division of Molecular and Clinical Medicine, University of Dundee, Dundee, UK
²Nottingham Digestive Diseases Centre, University of Nottingham, Nottingham, UK
³The Robertson Centre for Biostatistics, University of Glasgow, Glasgow, UK
⁴School of Pharmacy, University College London, London, UK
⁵Centre for Academic Primary Care, School of Medicine, University of Nottingham, Nottingham, UK
⁶Salus Alba, Glasgow, UK
⁷Institute of Cardiovascular Sciences, University of Birmingham, Birmingham, UK
⁸Academic Primary Care, University of Aberdeen, Aberdeen, UK

*Corresponding author i.s.mackenzie@dundee.ac.uk

Members of the ALL-HEART study group are listed in Appendix 1.

Disclosure of interests

Full disclosure of interests: Completed ICMJE forms for all authors, including all related interests, are available in the toolkit on the NIHR Journals Library report publication page at https://doi.org/10.3310/ATTM4092.

Primary conflicts of interest: Isla Mackenzie reports research grants from NIHR HTA for the submitted work, research grants from Menarini, EMA, Sanofi, HDR UK, British Heart Foundation and IMI and institutional consultancy income from AstraZeneca outside the submitted work, and personal income from AstraZeneca and Amgen outside the submitted work. She is a member of the NIHR Remote Trial Delivery Group representing the IMI Trials@Home consortium. Christopher Hawkey, Ian Ford, Nicola Greenlaw and Li Wei report research grants from NIHR HTA via University of Dundee to their institutions for the submitted work. Christopher Hawkey declares research funding from NIHR HTA for other work and consultancy income from Kallyopie. Amy Rogers declares membership of the Trial

Steering Committee for the DaRe2THINK (healthcare DAta for pragmatic clinical REsearch) trial, which is NIHR funded (NIHR HTA 13/02/80). Allan Struthers reports research grants from NIHR HTA for the submitted work and with the University of Dundee holds a European patent for the use of xanthine oxidase inhibitors in treating chest pain in angina pectoris. Alan Begg reports personal income from Novartis, Mylan, AstraZeneca, Bayer, Daiichi-Sankyo, Boehringer, Pfizer, Galderma, Zambon and Novo-Nordisk outside the submitted work and is a non-remunerated reviewer of cardiovascular applications to NIHR and the Honorary Treasurer of the Scottish Heart and Arterial disease Risk Prevention (SHARP) Society. Anthony Avery declares that he is a Member of the NIHR Programme Grants for Applied Research committee and is a NIHR Senior Investigator (NF-SI-0617-10076). He is also a recipient of research funding from non-commercial sources, mainly NIHR. He is National Clinical Director for Prescribing for NHS England. Jaspal Taggar declares research funding from NIHR and NHS England outside the submitted work, membership of a NICE guideline committee on management of atrial fibrillation, was a panel member for the NIHR RfPB award (2016–9), has received funding for research from the NIHR Clinical Research Network (CRN) East Midlands for the project: increasing engagement with and research opportunities for Eastern European and Ukrainian populations, and is a non-executive director for the Royal Derby and Burton Hospital NHS Trust. Andrew Walker declares personal income for consultancy from Abbvie, Akcea, Albireo, Alexion, Allergan, Amarin, Apsara, Arena, Astellas, AstraZeneca, Autolus, Bayer, Biocryst, Biogen, Biomarin, BMS, Boehringer Ingelheim, Calico, Celgene, Chiesi, Daiichi Sankyo, Diurnal, Elsai, Eli Lilly, Ferring, Galapagos, Gedeon Richter, Gilead, GSK, GW Pharma, Idorsia, Incyte, Intercept, Ionis, Ipsen, Janssen, Jazz, Jcyte, Kite Gilead, LEK, Leo Pharma, Les Laboratories Servier, Lundbeck, Merck (MSD), Merck-Serono, Mitenyi, Mundibiopharma, Mustang Bio, Mylan, Myovant, Norgine, Novartis, Novo Nordisk, Orchard, Paion, Pfizer, Pierre Fabre, PTC, RegenXBio, Rhythm, Sanofi, Santen, Sarepta, SeaGen, Shionogi, Sigmatec, SOBI, Takeda, Tanaya, UCB and Vertex outside the submitted work. Thomas MacDonald reports research grants from NIHR HTA for the submitted work, research grants from Menarini/Ipsen/Teijin, British Heart Foundation, Research Triangle Institute Barcelona, IMI and EMA outside the submitted work, institutional income for consultancy from AstraZeneca and speaker honoraria from Wellcome Trust outside the submitted work, membership of the steering committee of the HEAT study (NIHR HTA funded), membership of a data monitoring committee (Novartis) and is a trustee of the Scottish Heart and Arterial disease Risk Prevention (SHARP) Society. Filippo Pigazzani, Suzanne Duce, Rebecca Barr, Jennifer Dumbleton, Evelien Rooke, Jonathan Townend and Lewis Ritchie declare no competing interests.

Published March 2024 DOI: 10.3310/ATTM4092

Plain language summary

Allopurinol and cardiovascular outcomes in patients with ischaemic heart disease: the ALL-HEART RCT and economic evaluation

Health Technology Assessment 2024; Vol. 28: No. 18 DOI: 10.3310/ATTM4092

NIHR Journals Library www.journalslibrary.nihr.ac.uk

Plain language summary

What was the question?

The purpose of the ALL-HEART study was to determine whether giving allopurinol to people with ischaemic heart disease (also commonly known as coronary heart disease) would reduce their risk of having a heart attack, stroke or of dying from cardiovascular disease. Allopurinol is a medication usually given to patients with gout to prevent acute gout flares. It is not currently used to treat ischaemic heart disease.

What did we do?

We randomly allocated people aged over 60 years with ischaemic heart disease to take up to 600 mg of allopurinol daily (in addition to their usual care) or to continue with their usual care. We then monitored participants for several years and recorded any major health events such as heart attacks, strokes and deaths. We obtained most of the follow-up data from centrally held electronic hospital admissions and death records, making the study easier for participants and more cost-efficient. We asked participants in both groups to complete questionnaires to assess their quality of life during the study. We also collected data to determine whether there was any economic benefit to the NHS of using allopurinol in patients with ischaemic heart disease.

What did we find?

There was no difference in the risk of heart attacks, strokes or death from cardiovascular disease between the participants given allopurinol and those in the group continuing their usual care. We also found no difference in the risks of other cardiovascular events, deaths from any cause or quality-of-life measurements between the allopurinol and usual care groups.

What does this mean?

The results of the ALL-HEART study suggest that we should not recommend that allopurinol be given to people with ischaemic heart disease to prevent further cardiovascular events or deaths.

Health Technology Assessment

ISSN 2046-4924 (Online)

Impact factor: 3.6

A list of Journals Library editors can be found on the NIHR Journals Library website

Launched in 1997, *Health Technology Assessment* (HTA) has an impact factor of 3.6 and is ranked 32nd (out of 105 titles) in the 'Health Care Sciences & Services' category of the Clarivate 2022 Journal Citation Reports (Science Edition). It is also indexed by MEDLINE, CINAHL (EBSCO Information Services, Ipswich, MA, USA), EMBASE (Elsevier, Amsterdam, the Netherlands), NCBI Bookshelf, DOAJ, Europe PMC, the Cochrane Library (John Wiley & Sons, Inc., Hoboken, NJ, USA), INAHTA, the British Nursing Index (ProQuest LLC, Ann Arbor, MI, USA), Ulrichsweb[™] (ProQuest LLC, Ann Arbor, MI, USA) and the Science Citation Index Expanded[™] (Clarivate[™], Philadelphia, PA, USA).

This journal is a member of and subscribes to the principles of the Committee on Publication Ethics (COPE) (www.publicationethics.org/).

Editorial contact: journals.library@nihr.ac.uk

The full HTA archive is freely available to view online at www.journalslibrary.nihr.ac.uk/hta.

Criteria for inclusion in the Health Technology Assessment journal

Manuscripts are published in *Health Technology Assessment* (HTA) if (1) they have resulted from work for the HTA programme, and (2) they are of a sufficiently high scientific quality as assessed by the reviewers and editors.

Reviews in *Health Technology Assessment* are termed 'systematic' when the account of the search appraisal and synthesis methods (to minimise biases and random errors) would, in theory, permit the replication of the review by others.

HTA programme

Health Technology Assessment (HTA) research is undertaken where some evidence already exists to show that a technology can be effective and this needs to be compared to the current standard intervention to see which works best. Research can evaluate any intervention used in the treatment, prevention or diagnosis of disease, provided the study outcomes lead to findings that have the potential to be of direct benefit to NHS patients. Technologies in this context mean any method used to promote health; prevent and treat disease; and improve rehabilitation or long-term care. They are not confined to new drugs and include any intervention used in the treatment, prevention or diagnosis of disease.

The journal is indexed in NHS Evidence via its abstracts included in MEDLINE and its Technology Assessment Reports inform National Institute for Health and Care Excellence (NICE) guidance. HTA research is also an important source of evidence for National Screening Committee (NSC) policy decisions.

This manuscript

The research reported in this issue of the journal was funded by the HTA programme as award number 11/36/41. The contractual start date was in September 2013. The draft report began editorial review in Febuary 2023 and was accepted for publication in September 2023. The authors have been wholly responsible for all data collection, analysis and interpretation, and for writing up their work. The HTA editors and publisher have tried to ensure the accuracy of the authors' manuscript and would like to thank the reviewers for their constructive comments on the draft document. However, they do not accept liability for damages or losses arising from material published in this manuscript.

This manuscript presents independent research funded by the National Institute for Health and Care Research (NIHR). The views and opinions expressed by authors in this publication are those of the authors and do not necessarily reflect those of the NHS, the NIHR, the HTA programme or the Department of Health and Social Care. If there are verbatim quotations included in this publication the views and opinions expressed by the interviewees are those of the interviewees and do not necessarily reflect those of the NHS, those of the authors, those of the NHS, the NIHR, the HTA programme or the Department of Health and Social Care.

Copyright © 2024 Mackenzie *et al.* This work was produced by Mackenzie *et al.* under the terms of a commissioning contract issued by the Secretary of State for Health and Social Care. This is an Open Access publication distributed under the terms of the Creative Commons Attribution CC BY 4.0 licence, which permits unrestricted use, distribution, reproduction and adaptation in any medium and for any purpose provided that it is properly attributed. See: https://creativecommons.org/licenses/by/4.0/. For attribution the title, original author(s), the publication source – NIHR Journals Library, and the DOI of the publication must be cited.

Published by the NIHR Journals Library (www.journalslibrary.nihr.ac.uk), produced by Newgen Digitalworks Pvt Ltd, Chennai, India (www.newgen.co).